Author's response to reviews

Title: Exploring effectiveness of different health financing mechanisms in Nigeria; what needs to change and how can it happen?

Authors:
Obinna Onwujekwe (onwujekwe@yahoo.co.uk)
Nkoli Ezumah (ezumahnk@yahoo.com)
Chinyere Mbachu (chinyere23ng@gmail.com)
Felix Obi (obifelix@gmail.com)
Hyacinth Ichoku (hichoku@yahoo.com)
Benjamin Uzochukwu (bscuzochukwu@gmail.com)
Hong Wang (Hong.Wang@gatesfoundation.org)

Version: 3 Date: 26 Aug 2019

Author’s response to reviews:

Samuel Kotei Amon, MPH (Reviewer 2): I recently reviewed the attached manuscript and made a few recommendations which required authors' attention. Authors' have since revised the manuscript and described the changes made in a letter. My original points raised have largely been well addressed. However, there are still minor issues which need attention:

COMMENT: I suggest that the title of the article be revised again. Since the study is an exploratory one, I recommend that the 'Examining' be replaced with 'Exploring' to read "Exploring effectiveness of different health financing mechanisms in Nigeria; what needs to change and how can it happen?"

RESPONSE: The title has been revised accordingly and now reads, "Exploring effectiveness of different health financing mechanisms in Nigeria; what needs to change and how can it happen?"

COMMENT: The paper still require proofreading, as there are sentence construction issues that need attention

RESPONSE: All sections of the paper have been proofread. Poorly constructed sentences have been identified and rephrased.